Author's response to reviews

Title: Risk factors for ectopic pregnancy: a multi-center case-control study

Authors:

Cheng Li Mr. (downing@sjtu.edu.cn)
Wei-Hong Zhao Miss. (weihong-zhao@hotmail.com)
Qian Zhu Miss. (melodyqianzhu@126.com)
Shu-Jun Cao Prof. (shujun-cao@163.com)
Hua Ping Prof. (pinghua661106@126.com)
Xiaowei Xi Prof. (xixiaowei1966@126.com)
Guo-Juan Qin Miss. (xiaoguo19711001@163.com)
Ming-Xing Yan Miss. (yanxiaohei00000@163.com)
Duo Zhang Miss. (zd621@hotmail.com)
Jun Qiu Prof. (qjchildren736@qq.com)
Jian Zhang Prof. (zhangjian_ipmch@sjtu.edu.cn)

Version: 4
Date: 17 July 2015

Author's response to reviews: see over
Dear Dr Sherif Shazly and Mr Jason Pepito,

Thank you very much for having our revised manuscript (Risk factors for ectopic pregnancy: A multi-center case-control study. MS ID: 202386173158128) peer reviewed and felt it suitable for publication in BMC Pregnancy and Childbirth.

We are very much encouraged by the positive comments of the reviewers and editors, and grateful to you for recognizing the value of our work. According to the editors’ suggestions, we have provide email addresses for all authors, and provide a figure legend for figure 1, placed at the end of the main manuscript. Enclosed please find the cover letter containing detailed responses to the editorial comments, the clean revised manuscript, the revised manuscript with Track Changes, and the figure.

We are very pleased to see that the Editors/Reviewers’ comments have helped to make significant improvements to the quality of our manuscript. We hope that the revised manuscript will now be satisfactory to be accepted for publication in the BMC Pregnancy and Childbirth.

Thank you for your time and consideration. We are looking forward to hearing from you soon.

Kind regards,

Jian Zhang, M.D.
Department of Obstetrics and Gynecology, International Peace Maternity and Child Health Hospital, School of Medicine, Shanghai Jiaotong University
910 Hengshan Rd.
Shanghai, 200030
People’s Republic of China
E-mail: zhangjian_ipmch@sjtu.edu.cn
Tel: 86-21-64070434
Fax: 86-21-64073896
Response to the Editors

General Note:

The editorial comments were copied from BMC Pregnancy and Childbirth Editor Decision Letter without changes. Our responses addressing individual comments and critiques are intercalated below.

Editorial Comment:

1. Please provide email addresses for all authors.

Response:

According to the editors’ suggestions, we have provide email addresses for all authors.

(Please see title page)

2. Please provide a figure legend for figure 1, placed at the end of the main.

Response: Thanks for your careful observation. Figure legend for figure 1 has been provided at the end of the manuscript. (Please see page 21)